Author's response to reviews

Title: Causes, pattern and short-term outcome of chest injuries at Bugando Medical Centre, Mwanza, Tanzania

Authors:

Monafisha K Lema (monafishal@hotmail.com)
Phillipo L Chalya (drphillipoleo@yahoo.com)
Joseph B Mabula (dribmabula@yahoo.com)
William Mahalu (william.mahalu@yahoo.com)

Version: 2 Date: 13 August 2010

Author's response to reviews:

Dear Editor

Journal of Cardiothoracic Surgery

Re: Submission of Manuscript for Publication

We would like to make corrections in Table 7